Author’s response to reviews

Title: Fatalism in breast cancer and performing mammography on women with or without a family history of breast cancer

Authors:

Maryam Molaei Zardanjani (m_molaei@nm.mui.ac.ir)

Mitra Savabi Esfahani (m_savabi@nm.mui.ac.ir)

Fariba Taleghani (Taleghani@nm.mui.ac.ir)

Version: 5 Date: 02 Aug 2019

Author’s response to reviews:

Dear Sir,

We appreciate the opportunity to revise our manuscript. A point by point response to comments is saved as letter 4 in supplementary material. We have provided a table included comments, changes and where these can be found (section, line and page).

In addition changed sentences are highlighted in main manuscript.

Sincerely Yours

Mitra Savabi Esfahani